Author’s response to reviews

Title: Understanding accreditation standards in general practice – a qualitative study

Authors:

Tina Due (tina.due@sund.ku.dk)
Thorkil Thorsen (tthorsen@sund.ku.dk)
Marius Kousgaard (marbro@sund.ku.dk)

Version: 2  Date: 04 Jan 2019

Author’s response to reviews:

Dear Dr. Zalika Klemenc-Ketis

We are pleased to hear that our manuscript is almost ready for publication in BMC Family Practice.

We have carefully considered your comments regarding the conclusion and revised the conclusion accordingly. Hence, we have sharpened the conclusion by carving out the essential message of the paper more clearly. Doing so we have also cut out a few parts, including the references (as suggested).

Best regards

Tina Drud Due